Author’s response to reviews

Title: Perinatal mortality by gestational week and size at birth in singleton pregnancies at and beyond term: a nationwide population-based cohort study

Authors:

Nils-Halvdan Morken (nils-halvdan.morken@kk.uib.no)
Kari Klungsøyr (Kari.Melve@isf.uib.no)
Rolv Skjærven (Rolv.Skjærven@igs.uib.no)

Version: 4  Date: 5 May 2014

Author’s response to reviews:

To the Editor-in-chief,
BMC Pregnancy and Childbirth
May, 2014

Dear Sir,

Thank you for giving us the opportunity to comment on the posed comment and submitting a revised version of the paper entitled: “Perinatal mortality by gestational week and size at birth in singleton pregnancies at and beyond term: a nationwide population-based cohort study” for your consideration.

Editor’s comment:
"Remove the suffixes from the title page"
Response: We have removed the suffixes from the title page.

Regards,

Nils-Halvdan Morken M.D.,PhD Kari Klungsøyr M.D., PhD Rolv Skjærven Ph.D.

The corresponding author is:
Nils-Halvdan Morken
Department of Global Public Health
University of Bergen
5018 Bergen
Norway
Telephone: + 4792257114
E-mail: nils-halvdan.morken@kk.uib.no